Author’s response to reviews

Title: Feasibility of Community Based HIV Self-Screening in South Africa: A Demonstration Project

Authors:

Limakatso Lebina (lebinal@phru.co.za)
Ntombexolo Seatlholo (Seatlholop@phru.co.za)
Noah Taruberekera (ntaruberekera@psi.org)
Mopo Radebe (mopo@sfh.co.za)
Anthony Kinghorn (kinghorna@phru.co.za)
Tessa Meyer (tessa@sfh.co.za)
Miriam Mhazo (mariam@sfh.co.za)
Kennedy Otwombe (otwombek@phru.co.za)
Khuthadzo Hlongwane (hlongwanek@phru.co.za)
Ashley Ringane (ringanea@phru.co.za)
Nthabiseng Koloane (koloanen@phru.co.za)
Mbali Nkuta (www.nkutam@phru.co.za)
Nkhensani Nkhwashu (NkhensaniN@foundation.co.za)
Thato Farirai (ThatoF@foundation.co.za)
Patience Kweza (PatienceK@foundation.co.za)
Thato Chidarikire (www.matshabathato44@gmail.com)
Simukai Shamu (SimukaiS@foundation.co.za)
Tendesayi Kufa (TendesayiKC@nicd.ac.za)
Adrian Puren (adrianp@nicd.ac.za)
Neil Martinson (martinson@phru.co.za)
Author’s response to reviews:

Dear Reviewers and Editors
Thank you for reviewing the manuscript. We have attached a detailed letter on each of the comments.